Author's response to reviews

Title: Planning and problem-solving training for patients with schizophrenia: A randomized controlled trial

Authors:

Kathlen Rodewald (katlhen.rodewald@med.uni.heidelberg.de)
Mirjam Rentrop (mirjam.rentrop@med.uni-heidelberg.de)
Daniel V. Holt (daniel.holt@psychologie.uni-heidelberg.de)
Daniela Roesch-Ely (daniela.roesch@med.uni-heidelberg.de)
Matthias Backenstrass (m.backenstrass@klinikum-stuttgart.de)
Joachim Funke (joachim.funke@psychologie.uni-heidelberg.de)
Matthias Weisbrod (matthias.weisbrod@kkl.srh.de)
Stefan Kaiser (stefan.kaiser@puk.zh.ch)

Version: 4 Date: 20 April 2011

Author's response to reviews: see over
Dear Biomed Central Team,

herewith we submit the manuscript with the requested formatting changes. We briefly list the changes below:

- The numbering of the tables has been corrected as requested.
- All highlighting has been removed from the manuscript.
- In the tables all cell merging across rows has been removed.
- White borders of the figures have been cropped.

With best regards,

Stefan Kaiser